Author’s response to reviews

Title: Oncofertility healthcare circuit and networks worldwide: what are the issues?

Authors:

Kathleen MELAN (kathleen.melan@gmail.com)
Frederic AMANT (frederic.amant@uzleuven.be)
Jacqueline Veronique-Baudin (Jacqueline.Baudin@chu-fortdefrance.fr)
Clarisse JOACHIM (clarisse.joachim@chu-fortdefrance.fr)
Eustase JANKY (eustase.janky@chu-guadeloupe.fr)

Version: 2 Date: 22 Dec 2017

Author’s response to reviews:

BCAN-D-17-00300R2

Fertility preservation healthcare circuit and networks in cancer patients worldwide:

What are the issues?

BMC Cancer

1. In order to offer readers a chance to fully assess your methods and findings, your manuscript needs to be copyedited. We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central has a new in house editing service. The new editing tool can provide both scientific and language editing: http://authorservices.springernature.com/

The manuscript has been thoroughly revised by a native English-speaking medical writer. We hope that the quality of the English will now be considered acceptable.

2. On uploading your revisions, please remove all tracking/highlighting and instead upload a single clean version of the manuscript.
Thank you for bringing this to our notice. We have prepared a clean version as requested.

3. Please add a 'List of abbreviations' after the 'Conclusion'.

We have included a list of abbreviations after the Conclusion.

4. Please place 'List of figures and tables' after the 'References' and remove the section 'References included in figures and tables'.

This has been corrected and re-formatted as requested.

5. Please remove all section numbering.

All section numbering has been removed.

6. Under Funding (within declarations) - please add your reference number and the phrase: 'The funding body had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript'.

Thank you for bringing this to our notice. We have made the requested correction. The funding mentioned is a PhD scholarship awarded by the local government authorities, and has no specific reference number. Therefore, no funding reference number has been indicated.

7. Please remove the figure titles and captions from the actual figures.

All captions have been removed from the figures.